Dear Professor Ma,

We thank you for the opportunity to resubmit our manuscript to the World Journal of Gastroenterology and hope that it is suitable for publication. Also, we are grateful to the reviewer for the comments on the original version of our manuscript. We have carefully studied the items they have raised. The text in the manuscript has been revised in line with the reviewers’ suggestions essentially improving the manuscript.

Name of journal: World Journal of Gastroenterology

ESPS Manuscript NO: 93859

Manuscript Type: Letter to Editor

Title: Is endoscopic ultrasound a promising technique in the diagnosis and treatment of liver diseases?

All Author List: Enver Zerem, Željko Puljiz, Boris Zdilar, Suad Kunosić, Admir Kurtcehajic and Omar Zerem

With respect,

Corresponding author:

Enver Zerem (Number ID: 02440510) on behalf of all authors

Answers to Reviewers Comment:

All changes in the revised manuscript are marked in such a way that new text is colored in red and deleted in light blue.

Reviewer Comment

Title: the title properly explains the objective of the article. Based on his own clinical work experience, the author systematically expounds the application value of EUS, which has certain clinical practical value. Conclusion: In general: Good. The article needs further polishing

Answers to Reviewers Comment:

Since the reviewer did not mention specific remarks on the content of the text, we understood that the reviewer was suggesting additional language corrections.
Accordingly, we asked a native English speaker, who reviewed the text in detail once again and made additional corrections (new words in the text are colored red and deleted words from the first version of the text are colored in light blue).